

How Does Urbanization Affect Citizens' Pro-Environment Behavior? A Hierarchical Analysis of the Chinese General Social Survey

Wen Qing¹*, Yibao Wang¹*, Yanbo Zhang², Jingwen Han³ and Cuixi Li¹

¹School of Public Policy and Management, China University of Mining and Technology, Xuzhou, China, ²School of Management Engineering and Business, Hebei University of Engineering, Handan, China, ³School of Economics and Management, China University of Mining and Technology, Xuzhou, China

Many studies have focused on the pressure of urbanization on resources and environment, but few have explored the positive effect of urbanization on reducing environmental pressure from the perspective of urbanization affecting pro-environment behavior (PEB). To systematically explore the impact of urbanization with different dimensions on PEB, this study establishes a hierarchical theoretical framework and conducts empirical test based on the data of Chinese General Social Survey (CGSS) and China City Statistical Yearbook (CCSY) as well as a hierarchical linear model (HLM). The results reveal the following: First, urbanization can directly influence PEB and the four dimensions of urbanization (economic, social, spatial and population urbanizations) can influence PEB in different manners. Economic urbanization can positively influence private PEB; social urbanization exerts a slightly positive impact on PEB; spatial urbanization has a notably negative impact on PEB while population urbanization has no significant effect on PEB. Second, Urbanization also indirectly affects PEB by influencing individual characteristic variables. Social and economic urbanizations positively moderate the relationship between social interaction and PEB. The research results illustrate the relationship between urbanization and PEB, and can guide how to promote PEB in the process of urbanization.

Keywords: urbanization, pro-environment behavior, social interaction, environmental knowledge, hierarchical linear model, moderating effect

1 INTRODUCTION

Urbanization is the surest way toward modernization and a powerful engine for high-quality economic development. Since the reform and opening up, China has been experiencing an urbanization process with the largest scale at the highest speed in the world history, and has made remarkable achievements. According to the data from the National Bureau of Statistics, the urbanization rate of China's permanent residents increased from 17.9% in 1978 to 63.89% in 2020, an average annual growth rate of 1.09%, far higher than the global average over the same period (National Bureau of Statistics, 2021). Although urbanization promotes economic prosperity, it also causes great consumption of resources and energy and environmental pollution (Han et al., 2018; Jin et al., 2018; Zhang et al., 2021a; Zhang et al., 2021b). Statistics revealed that in the past decade, China ranked around 120th in the Global Environmental Performance Index (121th among 163 countries in the year of 2010 and 120th among 180 countries in the year of 2020), falling behind the average

OPEN ACCESS

Edited by:

Zhonghua Gou, Wuhan University, China

Reviewed by: Ramakrishnan

Jegan Ramakrishnan, Griffith University, Australia Gui Jin, China University of Geosciences, China

*Correspondence:

Wen Qing qingwen3509@cumt.edu.cn Yibao Wang wangyb@cumt.edu.cn

Specialty section:

This article was submitted to Environmental Economics and Management, a section of the journal Frontiers in Environmental Science

> Received: 21 February 2022 Accepted: 22 April 2022 Published: 10 May 2022

Citation:

Qing W, Wang Y, Zhang Y, Han J and Li C (2022) How Does Urbanization Affect Citizens' Pro-Environment Behavior? A Hierarchical Analysis of the Chinese General Social Survey. Front. Environ. Sci. 10:880852. doi: 10.3389/fenvs.2022.880852 level of countries with the same income (Wendling et al., 2020). Faced with severe environmental problems, the Chinese government attaches great importance to environmental protection and made strategic plans to promote ecological civilization for a beautiful China (Jin et al., 2020). The 14th Five-Year Plan for China's National Economic and Social Development (2021-2025) also clearly proposes to accelerate green and low-carbon development, continuously improve environmental quality, and integrate the concepts of "ecological civilization, greenness, low-carbon and economical and intensive utilization of resources" into the process of urbanization (Agency, 2021). However, the solution to environmental problems and the realization of green, lowcarbon and high-quality urbanization not only depend on the administrative government, but also require extensive and effective participation of the public. Therefore, it is of theoretical value and practical significance to study the impact of urbanization on pro-environment behavior (PEB) and how to promote PEB in the process of urbanization.

Previous researches have analyzed the pressure of urbanization on resources and environment, but few explored the positive effect of urbanization on environmental governance, especially its impact on private PEB. First, studies on PEB failed to discuss the relationship between urbanization and PEB. Previous studies on PEB mainly focused on individual characteristics, such as attitude and awareness (Azucena Vicente-Molina et al., 2013; Gifford and Nilsson, 2014), ignoring macro factors (Wang and Han, 2016; Hong and Park, 2018; Hao and Song, 2020), especially the impact of urbanization on private PEB. Even if some scholars have studied the impacts of population density and urban population ratio on public pro-environment awareness and PEB (Franzen and Meyer, 2010), they merely took urbanization as a control variable, rather than a core explanatory variable, to explore the causal relationship between the two. Second, most previous studies reflected the process of urbanization in China through population urbanization (POU). Although population migration from rural areas to cities is the basic form of urbanization, in essence, urbanization is not a simple population agglomeration or urban sprawl, but a complex process that triggers the overall evolution of social and economic structure (Chen et al., 2020; Yao et al., 2021). In other words, previous studies failed to realize that urbanization is a multi-faceted process involving population agglomeration, economic development, formation of urban carriers with modern civilization, as well as improvement of infrastructure, public services and residents' living quality (Liu et al., 2019; Ren and Yu, 2021). Different aspects of urbanization may have different impacts on PEB. Finally, previous researches have not discussed the moderating effect of urbanization on the relationship between individual factors and PEB. The Attitude-Behavior-Condition (ABC) model proposes that external factors can affect PEB both directly and indirectly through interaction with individual factors (Guagnano et al., 1995; Steg and Vlek, 2009). PEB can be more accurately predicted by considering the direct and indirect effects of external factors. However, previous researches did not discuss the cross-level moderating effect of external factors on PEB, let alone the moderating effect of urbanization on the relationship between individual factors and PEB.

To make up for the above research defects, this study establishes a hierarchical theoretical framework. The impact of urbanization on

PEB is empirically analyzed with the data from Chinese General Social Survey (CGSS) 2013 and China City Statistical Yearbook (CCSY) 2013 by the hierarchical linear model (HLM). Special attention is paid to the impacts of different dimensions of urbanization on PEB. It also examines how urbanization moderates the relationship between individual factors and PEB. This research is of both theoretical and practical significance. First, the study expands the scope of relevant researches on the influencing factors for PEB, and grasps the complex relationship between urbanization and PEB by considering the different dimensions of urbanization. Previous researches on the influencing factors for PEB were mainly focused on the micro-level factors such as attitude, awareness and values (Azucena Vicente-Molina et al., 2013; Gifford and Nilsson, 2014), yet the macro-level factors were ignored, especially the impact of urbanization on PEB. This study investigates urbanization and PEB and finds that urbanization is an important influencing factor for PEB. The influences of different dimensions of urbanization on PEB differ greatly. Specifically, economic urbanization (ECU) significantly promotes private PEB, but has no significant impact on public PEB; social urbanization (SOU) exerts a limited positive impact on PEB; spatial urbanization (SPU) notably restricts PEB; population urbanization (POU) has no significant impact on PEB. The results reveal the complicated relationship between urbanization and PEB. Second, the study systematically analyzes how factors at city and individual levels influence PEB from the perspective of hierarchy; especially, it discusses the direct and indirect impacts of urbanization on PEB, providing an in-depth understanding of the relationship between the two. Third, since promoting the sustainable development of urbanization has become a common challenge, the theoretical framework and research findings proposed in this study are also informative for other countries. Changes in PEB will have a sustained and far-reaching impact on the reduction of urban environmental pressure. Future researches should pay more attention to the complex impact of urbanization on PEB, and consider the cross-level moderating effect. Finally, this study provides empirical evidence for PEB in urbanization. This helps China to steadily pursue green, low-carbon, sustainable and highquality urbanization.

This paper is structured in six sections. Section 2 reviews the relevant literature on urbanization and PEB, and puts forward research hypotheses. Section 3 introduces the research design, including data, variables and models. Section 4 gives the empirical analysis results. Section 5 discusses the main research results. Finally, Section 6 presents the research conclusions and some policy recommendations.

2 LITERATURE AND HYPOTHESES

2.1 Urbanization and Pro-Environment Behavior

Urbanization refers to the process of rural population migrating to cities, which is driven by the economic structure transformation, especially the transformation of industrial structure from an agriculture-based one to a non-agriculturebased one. This process is accompanied by multi-dimensional revolutions in economic growth mode, social organization structure and residents' lifestyle (Davis and Henderson, 2003). With the advancement of urbanization in China, new urbanization is gradually known to the public. New urbanization is not merely the urbanization of population, but a four-sphere integrated urbanization of population, economy, space and society (Liu et al., 2019; Ren and Yu, 2021; Yu, 2021). It is people-centered, pays attention to improving the quality of urbanization and strives to achieve equalization of public services.

PEB refers to the behavior actively taken by individuals in daily life to improve environmental conditions and environmental quality (Ates, 2020). There are different types of PEB: Stern and Paul (2000) were the first to classify them into three types: 1) radical PEB, such as actively participating in social movements; 2) public non-radical PEB, such as signing petitions for environmental issues and participating in processions; 3) private PEB, such as green consumption and energy-saving housing. Afterwards, many scholars combined the first two types of behaviors and further divided behaviors from two dimensions: private PEB and public PEB (Hunter et al., 2004; Hadler and Haller, 2011; Xiao and McCright, 2014). There are also other methods of classification, such as environmental activism and environmentally friendly behavior (Tindall et al., 2003), political PEB and protective PEB (Dalton, 2015). Different classification methods correspond to different research conclusions on the general pattern of environmental behavior (Xiao and McCright, 2014). To accurately investigate the influence of urbanization on different types of PEB, this study adopts the classification method of Hunter et al. (2004) and Wang and Han (2016), and divides PEB into private PEB and public PEB.

Urbanization, an inevitable process for economic development, has caused environmental pollution, but it may also improve PEB. For example, from a macro perspective, the affluence hypothesis holds that the wealth of a country or a region is positively correlated with its public pro-environment awareness and PEB (Diekmann and Franzen, 1999). Therefore, urbanization may promote PEB while contributing to economic development. Urbanization may also promote PEB by jointly improving the public services and the living quality. Studies have proven that basic education and social security can significantly promote PEB (Fan et al., 2018). From a micro perspective, urbanization contributes to education; a good education can promote PEB (Monier et al., 2009; Azucena Vicente-Molina et al., 2013; Xiao and McCright, 2014; Meyer, 2015; Zheng et al., 2019) by providing the access to pro-environment knowledge and delivering a higher level of pro-environment awareness. Therefore, theoretically, urbanization may have a complex impact on PEB.

2.1.1 Population Urbanization and Pro-Environment Behavior

Population urbanization (POU) refers to the process of population agglomeration from rural areas to urban areas. It is the product of social productivity development in a certain stage, and also the most typical phenomenon and result of urbanization. This agglomeration is not only an increase in urban population, but also a growth of urban population density, an increase in the number of employed people in the secondary and tertiary industries and a continuous improvement of urban population quality. POU has a complex impact on PEB. First, in the process of POU, the rapid growth of urban population intensifies the demand for various resources and energy, which will cause more pollutants and greenhouse gases, bringing greater pressure to the environment (Lin et al., 2017). Studies suggest that a high population density may highlight potential conflicts between economic growth and environmental quality, so people in densely populated countries may be more concerned about the environment (Franzen and Meyer, 2010). Consequently, countries or regions with higher degrees of POU may pay more attention to environmental issues, including greater desire to participate in environmental protection (Franzen and Meyer, 2010).

Second, from the perspective of population migration, migrants may also improve their PEB by learning and interacting with local urban residents. For example, studies hold that migrants can acquire more environmental knowledge (EK) and improve their PEB by actively integrating into local community and contacting local residents (Pfeffer and Stycos, 2002). However, some studies have different views. Hunter (2000) found that migrants and local residents differ insignificantly in PEB. Ng (1998) believes that migrants exert a weaker impact on the environment of their destination, compared with social institutional factors, such as economy, political structure and religious belief. In general, few literatures explored the impact of POU on PEB, and they have not arrived at a consistent conclusion. In the context of rapid urbanization in China, compared with people in cities with lower urbanization levels, people in cities with higher urbanization levels are more willing to participate in environmental protection. This study proposes the following hypotheses:

H1a: POU can promote private PEB.

H1b: POU can promote public PEB.

2.1.2 Economic Urbanization and Pro-Environment Behavior

Economic urbanization (ECU) promotes the comprehensive development of urban economy. For one thing, ECU is demonstrated by the agglomeration of production factors in cities, resulting in economies of scale and improving the income of urban residents. For another, ECU also contributes to the upgrading of industrial structure, namely, the transformation from an agriculture-based on to an industrybased one, then to a tertiary industry-based one. The affluence hypothesis (Diekmann and Franzen, 1999) believes that economic development can promote public pro-environment awareness and PEB. It holds that first of all, environmental quality is a normal commodity whose demand rises with prosperity (Markandya et al., 1979). Therefore, economic growth leads to an increase in public demand for environmental quality. Second, budgetary constraints can be released only when individual wealth increases, making it possible for individuals to invest more resources to improve environmental quality. In short, as economy develops, the public become more affluent; then their demand and ability to improve environmental quality also improve.

Relevant studies support the affluence hypothesis. For example, some studies find that environmental concerns at the individual level are positively correlated with a country's economic affluence measured by gross domestic product per capita (GDP) (Summers and VanHeuvelen, 2017; Welsch and Kühling, 2017). There is a clear relationship between economic development and political PEB or protective PEB (Dalton, 2015). ECU is manifested by the growth of economic output value as well as industrial transformation and upgrading. The proportion of industrial structure is closely related to the development of a region. In general, the higher the degree of industrialization in a region is, the more serious the environmental problems might be. The problems can well stimulate the environmental concern and PEB of citizens (Chen et al., 2013). Therefore, people in cities with higher levels of ECU are more willing to participate in environmental protection than those in cities with lower levels of ECU.

H2a: ECU is conducive to private PEB.

H2b: ECU is conducive to public PEB.

2.1.3 Spatial Urbanization and Pro-Environment Behavior

Spatial urbanization (SPU) is a phenomenon reflected on the carrier with ECU and POU, which is characterized by high-level organization of environment and agglomeration of material facilities (Ding and Zhao, 2011; Kuang et al., 2020). The so-called environmental organization refers to the artificial transformation of the natural environment, reflected in the existence of buildings and structures, making the environment suitable for human needs. The agglomeration of material facilities is another physical characteristic of urbanization, which is manifested in the high-density distribution of various workplaces, residences, transportation, recreation and other facilities in urban areas. Studies have shown that the construction of high and dense buildings promotes efficient and intensive use of land in urbanization, and provides amenities and services to nearby residents. These not only bring a more convenient urban life in which walking is preferred, but also stimulate social interaction, social capital and community awareness (Montgomery, 1998; Cao et al., 2019). However, the development of high-density urban space also results in a series of problems such as congestion, lack of privacy and shortage of resources and services (Conway and Adams, 1977; Wang and Chien, 1998).

Studies have highlighted the importance of infrastructure for PEB. For example, the number (Derksen and Gartrell, 1993; Kuang and Lin, 2021), location (O'Connor et al., 2010) and availability (Timlett and Williams, 2009; Bernstad, 2014) of recycling facilities are all important factors for waste recycling. Besides, traffic conditions and urban greenness have a significant impact on public riding behavior, and people who ride bicycles prefer roads or streets with good connectivity, fewer vehicles (Sun et al., 2017), well cycle path condition (Etminani-Ghasrodashti et al., 2018) and more street greenery (Wang et al., 2020). In addition, optimizing public transport services and policies, such as offering cheap or free bus tickets, can also encourage people with high pro-environment awareness to minimize the use of private vehicles (Fan et al., 2018). As confirmed by these examples, SPU plays an important role in changing PEB. As China's urbanization advances, infrastructure has been improved significantly, but problems such as poor infrastructure quality, low convenience and extensive operation and management become increasingly prominent, which may constrain PEB (Fan et al., 2018). The following hypotheses are proposed:

H3a: SPU may constrain private PEB.

H3b: SPU may constrain public PEB.

2.1.4 Social Urbanization and Pro-Environment Behavior

Social urbanization (SOU) is the ultimate goal of urbanization whose main task is to improve the public services, such as medical service, education and social security, and to improve the living quality of urban residents. SOU may jointly promote PEB by improving the public services and living quality. Basic public services are the most relevant to public interest and a comprehensive variable affecting public behavior in daily life. Studies have confirmed that basic education and social security levels play a significant role in promoting PEB (Fan et al., 2018). In addition, as public service and life quality in urban areas improve, people's life improves in both material and cultural terms. According to the post-materialism theory, when the material needs are met to a certain extent, people begin to care about specific social issues, such as environmental issues. Specifically, according to Inglehart (1995), the measure of individual value is determined during primary socialization. People experience primary socialization in good socioeconomic conditions and form post-materialistic values in the context of economic growth, national security and widespread employment opportunities. As the social wealth accumulates, the public's transformation from the original "materialistic values" to "post-materialistic values" contributes to their concern for the environment and improves public environmental participation and support. Different from "materialistic values" which concern basic material needs, "post-materialistic values" emphasize living quality, self-expression and environmental protection. Relevant studies also provide supportive evidence. For example, Gelissen (2007) found that on average, the public with higher postmaterialist tendencies were more willing to make economic sacrifices for environmental protection after investigating the degree and difference of public support for environmental protection in 50 countries. Therefore, people in cities with higher levels of SOU are more willing to participate in environmental protection than those living in cities with lower levels of SOU. This study proposes the following hypotheses:

H4a: SOU facilitates private PEB.

H4b: SOU facilitates public PEB.

2.2 Cross-Level Moderation of Urbanization on Pro-Environment Behavior

As is pointed out by the ABC model, external factors can affect PEB both directly and indirectly through interaction with individual factors (Steg and Vlek, 2009). As an important moderating mechanism, external factors can strengthen or weaken the impact of individual factors on PEB (Guagnano et al., 1995; Stern, 2000). Researches that merely focus on the main effect are misleading (Ertz et al., 2016) because they ignore the moderating effect. In contrast, researches considering the direct and indirect effects of external factors can improve the explanatory ability of the theoretical model to predict PEB. Therefore, this research also focuses on the cross-level moderation of urbanization between individual factors and PEB.

First, some studies have emphasized the close relationship between social interaction (SI) and PEB, and found that the public not only adjust their PEB through the direct observation of the behavior of others, but also obtain pro-environment information and knowledge through SI, thus exerting a positive impact on their own behavior (Zheng et al., 2019). Compared with cities with low urbanization levels, cities with high urbanization levels are of more communication and more opportunities for interaction between different occupational groups, which is conducive to promoting SI (Sato and Zenou, 2015). Therefore, the impact of SI on PEB may vary with the level of urbanization.

H5a: Urbanization positively moderates the relationship between SI and private PEB and public PEB.

Second, the important predictive effect of EK on PEB has been confirmed by many researches. Levine and Strube (2012) found that general ecological knowledge can well predict PEB. Liu et al. (2020) and Wu et al. (2022) concluded that EK can promote PEB by shaping environmental attitudes and behavioral intentions. A recent study also revealed that changes in PEB are caused by EK (internal factor) and socio-economic factors (external factors) (Amoah and Addoah, 2021). Cities with higher urbanization levels correspond to better education and more channels to contact with the network media, which is helpful to acquire more EK. Therefore, the impact of EK on PEB may be moderated by the level of urbanization. Then the following hypothesis is proposed:

H5b: Urbanization positively moderates the relationship between EK and private PEB and public PEB.

2.3 Other Factors Affecting Pro-Environment Behavior

Education exerts superimposed effects on PEB. First, education has the social function of values, enhances people's proenvironment awareness, so that people are actively involved in environmental protection activities; second, individuals with higher education levels tend to have higher incomes and are more concerned about environmental issues. Empirical studies have also confirmed that education is an important factor in predicting pro-environment concern and behavior. For example, Meyer (2015) found that education makes individuals more concerned about social welfare and thus they act in a more environmentally friendly manner. Monier et al. (2009) found a positive correlation between education and the purchase of green food. Xiao and McCright (2014) found that education plays a significant role in promoting both private PEB and public PEB. Besides, as disclosed in many studies, there are gender differences in PEB. Compared with men, women are more frequently involved in private PEB, such as recycling and green consumption (Macias and Williams, 2016; Strapko et al., 2016; Pisano and Lubell, 2017; Ahmad et al., 2021; Fan et al., 2021). A common argument is that women have stronger proenvironment values, beliefs and attitudes than men (Xiao and Dunlap, 2007; McCright and Xiao, 2014). Some studies hold different views that women's lower levels of EK than men hinder their participation in private environmental protection (Xiao and Hong, 2018). Most studies consider that age is positively correlated with PEB, that is, the elderly are more involved in PEB than the young (Shen and Saijo, 2008; Wang et al., 2021). With the increase in age, people may increasingly seek the meaning of self-transcendence and the goal of pro-society. Therefore, the practice of PEB may become a way for the elderly to impart this wisdom and remain active (Wang et al., 2021).

The environmental pollution-driving hypothesis holds that serious environmental pollution can raise the public's proenvironment awareness, and then the public would take various PEBs. This hypothesis has been partially supported by empirical evidence (Inglehart, 1995; Wang and Han, 2016). However, some scholars believe that objective environmental pollution differs from public perception of environmental pollution. Usually, pollutants cannot be directly perceived by the public and therefore fail to influence their behaviors (Hyslop, 2009). Some scholars further proposed that this hypothesis should take public perception of environmental pollution into consideration (Franzen and Meyer, 2010; Mi et al., 2021). Therefore, in this study, the impacts of objective and subjective perceptions of environmental pollution on PEB are involved.

Based on the above analysis, a theoretical framework is established (Figure 1). This research focuses on the direct effects of the four dimensions of urbanization on PEB and its cross-level moderating effect between individual factors and PEB.

3 DATA AND METHODS

The data in this study are from two databases. 1) The micro data at the individual level are from CGSS 2013, conducted by the China Survey and Data Center of Renmin University of China, which is one of the most comprehensive social surveys in China. The survey adopted a four-level stratified sampling scheme. A total of 100 counties (or districts) were selected nationwide. Four communities or rural areas were randomly selected in each county (or district), where 25 families were randomly investigated. One person from each family was randomly selected for face-to-face interview. The survey included 28 provincial administrative units, excluding China's Hong Kong SAR, Macao SAR, Tibet Autonomous Region, Xinjiang Uygur Autonomous Region and Hainan Province, with a total sample size of 11,438 people, and it was representative of the characteristics of the country. As the research object was urban residents, the final effective sample size for this study was 6,728 people distributed in 75 cities across China after rural samples and invalid data were excluded. 2) The data at the city level are from CCSY 2013, published annually by the National Bureau of Statistics of China, which is a compilation of statistical



TABL	TABLE 1 Statistics of PEB (unit: %).									
NO.	Environmental protection activities	Never	Occasionally	Often						
	or behaviors									
1	Garbage sorting	45.3	38.9	15.8						
2	Discuss environmental issues with your friends and relatives	42.1	47.3	10.6						
3	Bring your own shopping basket or shopping bag when purchasing daily necessities	18.8	35.4	45.8						
4	Reuse plastic packaging	14.5	30.3	55.2						
5	Donate for environmental protection	40.9	42.3	16.8						
6	Actively focus on environmental issues reported in radio, television and newspapers and environmental information	75.8	21.3	2.9						
7	Actively participate in environmental publicity and education activities organized by the government and organizations	70.5	24.2	5.3						
8	Actively participate in environmental protection activities organized by private environmental groups	78.2	18.7	3.1						
9	Self-funded forest or green space	85.8	11.1	3.1						
10	Actively participate in complaints and appeals that require environmental problems	87.5	10.4	2.1						

data that comprehensively reflects the economic and social development of China's cities. The data in the *CCSY 2013* consist of a series of indexes such as urban economic and social development and urban construction in 2012, covering all the 657 cities in China.

3.1 Variables and Operations

3.1.1 Dependent Variables

The dependent variable of this study is the private PEB. In 2013, CGSS measured PEB by 10 questions (**Table 1**). According to the analysis of Wang and Han (2016), this study divides PEB corresponding to 10 questions into two categories: private PEB and public PEB. Questions 1-4 and 6 belong to private PEB, which refers to the PEB in daily life. Questions 5 and 7–10 are public PEB, which refer to the non-radical actions with public's willingness to improve environmental quality. The given options are as follows: 1 = ``Never'', 2 = ``Occasionally'' and 3 = ``Often.'' With reference to the researches (Hadler and Haller, 2011; Wang and Han, 2016), the measurement classification of PEB was indexed. Take the private PEB index as an example: First, all

the values were accumulated and divided by 5 to obtain the average value X. Then, it was processed into percent $(X-1) \times (100/2)$. Finally, the private PEB index in the range of 0–100 was obtained. 0 represents the minimum private PEB while 100 represents the maximum. Public PEB was processed in the same way.

3.1.2 Independent Variables

3.1.2.1 City Level

Urbanization is a key variable in the research. As mentioned above, most previous studies measured the level of urbanization by the proportion of urban population which merely reflects the degree of population agglomeration. In fact, urbanization is not just the urbanization of population, but a complex process that causes the overall evolution of social and economic structure (Chen et al., 2020). In recent years, some scholars have established an index system of urbanization from various perspectives and drew a relatively consistent conclusion that the level of urbanization should be measured more comprehensively and scientifically from four dimensions,

TABLE 2	Evaluation	indexes	at the	urbanization I	evel.

System	First-level indications	Second-level indicators (Unit)	Variable	Positive/negative
Urbanization	Population urbanization (POU)	Proportion of urban population (%)	X ₁	+
		Urban population density (person/km ²)	X ₂	_
		Proportion of employed persons in secondary and tertiary industries (%)	X ₃	+
	Economic urbanization (ECU)	Per capital GDP (yuan)	X ₄	+
		Per capita local fiscal expenditure (yuan)	X ₅	+
		The proportion of tertiary industry in GDP (%)	X ₆	+
	Spatial Urbanization (SPU)	Proportion of urban built-up area in total area (%)	X ₇	+
		Green coverage rate of completed areas (%)	X ₈	+
		Number of buses per 10,000 people (Vehicle)	X ₉	+
	Social Urbanization (SOU)	Per capita consumption expenditure of urban residents (yuan)	X ₁₀	+
		The number of college students per 10,000 people (person)	X ₁₁	+
		Number of health technicians per thousand people (person)	X ₁₂	+
		Proportion of urban basic medical insurance participants (%)	X ₁₃	+

i.e., POU, ECU, SPU and SOU (Liu et al., 2019; Kuang et al., 2020; Yu, 2021). This research recognizes this view. Based on previous research results and scientific, systematic and feasible principles, an urbanization evaluation system consisting of four first-level indexes of POU, ECU, SPU and SOU as well as 13 second-level indexes was established (**Table 2**).

The specific explanation is as follows. First, the agglomeration of population from rural areas to urban areas is the most typical phenomenon and result of urbanization. This agglomeration is not only an increase in the urban population, but also a growth of urban population density and the number of people in the secondary and tertiary industries. Thus, the indexes to measure POU are X1-X3 (Liu and Lei, 2018). Second, the agglomeration of production factors in cities and towns has improved production efficiency, resulting in economies of scale, promoting the optimization and upgrading of industrial structure and improving the income of urban residents. Therefore, ECU is evaluated by economic development, industrial structure development and fiscal expenditure; the specific measurement indexes are X4-X6 (Shang et al., 2018; Yu, 2021). Third, SPU is the carrier of urbanization, and the advancement of urbanization is surely reflected in space, i.e., the external performance of urbanization in regional space, including the formation of urban carriers with modern civilization characteristics and the improvement of infrastructure such as traffic conditions. Hence, SPU is evaluated from urban coverage, green coverage in built-up areas and public transportation conditions. The specific measurement indexes are X7-X9 (Kuang et al., 2020; Yao et al., 2021). Fourth, SOU aims to promote the equalization of basic public services with medical service, education and social security at their core, and strives to improve public services, so that the public can enjoy the benefits brought by social resources and economic development. Therefore, SOU is evaluated from three aspects, namely medical treatment, education and social security, and the specific measurement indexes are X10-X13 (Liu et al., 2019; Kuang et al., 2020; Yu, 2021). This study incorporates POU, ECU, SPU and SOU into the model to explore how urbanization affects PEB. The measurement data of indexes come from CCSY 2013. As the

units of indexes are not the same, the second-level indexes were normalized. Ruxin (2017) suggested that the second-level indexes of urbanization could be assigned the same weight; then the first-level indexes were synthesized according to the equal weight, and the values of urbanization in four dimensions were obtained.

Environmental pollution index (*EPI*): With reference to the research of Fan et al. (2018), a comprehensive environmental pollution index was established which included industrial emissions of sulfur dioxide (SO₂), dust and fume and wastewater to evaluate environmental pollution. The measurement data are from *CCSY 2013*. First, to eliminate the influence of different dimensions, the indexes were normalized. Then the comprehensive score of environmental pollution index was calculated by the weighted average method; next, the index was divided by the territorial area of each city to eliminate the influence of different city scales.

3.1.2.2 Individual Level

Individual variables include EK and SI. According to previous researches (Zheng et al., 2019), the question "frequency of social entertainment activities with friends" could measure SI. The options were "almost every day," "once or twice a week," "several times a month," "nearly once a month," "several times a year," "once a year or less" and "never." The third and fourth options were merged, so were the fifth and sixth options; then their corresponding score assignments were 5, 4, 3, 2 and 1. A higher score suggested a higher social frequency of the respondents.

EK adopted a series of cognitive indexes of environmental problems. 1 point could be earned for correct answers while 0 point for wrong answers. Then the scores of indexes were accumulated to obtain the scores of EK (0–10). A higher score meant a higher level of EK.

Environmental pollution perception index (*EPPI*) was measured by respondents' perception of local environmental pollution. Furthermore, the variables were processed according to the severity assignment of the respondents' perceptions of air, water, noise, industrial waste, domestic waste and food pollution in the region, and with reference to the exponential processing

Variables		Definition	N	Mean	SD	Min	Max
Dependent	Private PEB	Continuous variable	6,728	48.18	23.13	0	100
variables	Public PEB	Continuous variable	6,728	11.88	17.19	0	100
Individual Level	Social interaction (SI)	Never = 1, Almost every day = 5	6,728	2.85	1.03	1	5
	Environmental Knowledge (EK)	Continuous variable	6,728	5.47	ean SD Min N 3.18 23.13 0 1 .88 17.19 0 1 .85 1.03 1 1 .47 2.73 0 1 .47 2.73 0 1 .17 16.52 17 16 .51 0.5 0 0 .36 4.34 0 26 .26 0.15 0.06 0 .39 0.16 0.04 0 .32 0.14 0.11 0	10	
	EPPI	Continuous variable	6,728	63.49	20.02	0	100
	Age	Continuous variable	6,728	47.17	16.52	17	97
	Gender	Male = 0, Female = 1	6,728	0.51	0.5	0	1
	Education	Continuous variable	6,728	10.36	4.34	0	19
City Level	POU	Continuous variable	75	0.26	0.15	0.06	0.91
	ECU	Continuous variable	75	0.28	0.18	0.07	0.92
	SPU	Continuous variable	75	0.39	0.16	0.04	0.75
	SOU	Continuous variable	75	0.32	0.14	0.11	0.8
	Environmental pollution index (EPI)	Continuous variable	75	0.14	0.14	0	0.98

method of the above dependent variables. The obtained EPPI was in the range of 0-100.

In addition, the research also controlled variables such as age, gender and education. The variables and definitions at the individual level are detailed in **Table 3**.

3.2 Model and Analysis Strategy

Data of individual-level PEB are nested within the city. Traditional regression analysis fails to solve the problem that data nesting violates the independent hypothesis of regression. However, the HLM can not only solve the problem facing data nesting, but also provide more macro-level information (Klein et al., 2000). Therefore, this study adopted the random intercept model in HLM to analyze the data. The private and public PEB were modeled respectively. Take private PEB as an example and the operations are as follows:

First, a null model without any explanatory variables was estimated; the sources of differences in PEB were decomposed into variances within and between cities; and individual-level explanatory variables were further added to the benchmark model on the basis that the intra-group correlation coefficient was obviously not 0. The null model is as follows:

Individual Level (Level 1):
$$Y_{ij} = \beta_{0j} + r_{ij}$$

City Level (Level 2): $\beta_{0j} = G_{00} + U_{0j}$,

where Y_{ij} , the dependent variable, is PEB score obtained by the *i*-th respondent in the *j*-th city; β_{0j} is the average value of PEB in the *j*-th city, and the variance of the error term r_{ij} represents the variation within the group; G_{00} represents the total average of PEB, and the variance of the error term U_{0j} is the variation between groups.

Subsequently, city-level variables and environmental pollution variables were included in the model, and the four measurement variables and environmental pollution variables of urbanization were included in the model to systematically examine the direct impact of urbanization on PEB. Finally, to explore the cross-level moderation of urbanization on individual factors and PEB, the four measurement variables of urbanization were further added to the coefficient equation of intercept and the two independent variables at individual level. The final complete model is as follows:

Individual level:

$$\begin{aligned} Y_{ij} &= \beta_{0j} + \beta_{1j} (SI)_{ij} + \beta_{2j} (EK)_{ij} + \beta_{3j} (EPPI)_{ij} + \beta_{4j} (Age)_{ij} \\ &+ \beta_{5j} (Gender)_{ij} + \beta_{6j} (Education)_{ij} + r_{ij} \end{aligned}$$

Contextual level:

$$\begin{split} \beta_{0j} &= G_{00} + G_{01} \, (POU)_{ij} + G_{02} \, (ECU)_{ij} + G_{03} \, (SPU)_{ij} \\ &+ G_{04} \, (SOU)_{ij} + G_{05} \, (EPI)_{ij} + u_{0j}; \\ \beta_{1j} &= G_{10} + G_{11} \, (POU)_{ij} + G_{12} \, (ECU)_{ij} + G_{13} \, (SPU)_{ij} \\ &+ G_{14} \, (SOU)_{ij} + u_{1j}; \\ \beta_{2j} &= G_{20} + G_{21} \, (POU)_{ij} + G_{22} \, (ECU)_{ij} + G_{23} \, (SPU)_{ij} \\ &+ G_{24} \, (SOU)_{ij} + u_{2j}; \\ \beta_{3j} &= G_{30} + u_{3j}; \\ \beta_{4j} &= G_{40} + u_{4j}; \\ \beta_{5j} &= G_{50} + u_{5j}; \\ \beta_{6i} &= G_{60}. \end{split}$$

In the complete model, β_{0j} shows the influence of city-level variables on the average PEB value of the *j*-th city. β_{1j} and β_{2j} are the estimates of intra-group relationships between SI and urbanization variables and between EK and urbanization variables, respectively. $G_{11} - G_{14}$ and $G_{21} - G_{24}$ reflect the cross-level moderation of urbanization on the relationship between individual factors and PEB. Data analysis was conducted with the aid of HLM software (version 6.08) which estimated the value of each parameter by the full maximum likelihood method.

The variance inflation factor (VIF) was adopted to examine the multi-collinearity of independent variables at the individual level. The test analysis shows that the VIF value lies in the range of 1.021–1.579, smaller than 10, indicating that there is no multi-collinearity between the variables at the individual level. In addition, the correlation coefficient value between city-level variables is in the range of 0.271–0.581.

TABLE 4 Impact of urbanization on private PEB: the results of the HI	LM.
--	-----

	Model 1	Model 2	Model 3	Model 4
Fixed Effect	Coef.	Coef.	Coef.	Coef.
Intercept	44.295***	44.275***	41.863***	41.083***
Individual Level				
SI		0.756***	0.756***	1.095***
EK		1.802***	1.802***	2.028***
EPPI		0.050**	0.050**	0.051**
Age		0.144***	0.144***	0.146***
Gender		-4.852***	-4.852***	-5.000***
Education		1.022***	1.022***	0.994***
City Level				
POU			-13.133	-13.129
ECU			17.621*	17.618*
SPU			-15.754*	-15.757*
SOU			12.391	12.391
EPI			23.099**	23.101**
Cross-Level Interaction	on			
POU × SI				3.640
ECU × SI				-7.476***
SPU × SI				-1.930
SOU × SI				5.116**
POU × EK				0.405
ECU × EK				-0.445
SPU × EK				-0.308
SOU × EK				-2.116
Random Effect	Variance	Variance	Variance	Variance
Level 2 effect	70.444***	70.061***	54.601***	54.626***
Level 1 effect	442.539	394.93	394.946	393.316

Note: *p < 0.1, **p < 0.05, ***p < 0.01; City(N) = 75, Individual(n) = 6,728.

According to Tsui et al. (1995), the critical value of correlation level for serious collinearity is generally over 0.75. Therefore, there is no serious multi-collinearity between the variables at the city level in this study. In addition, to prevent the multi-collinearity of interaction terms and original variables, the city-level urbanization variables and the individual-level SI and EK variables were centralized.

4 RESULTS

4.1 Descriptive Statistics

Table 3 gives the descriptive statistics. The average value of private PEB index is 48.18, while that of public PEB index is 11.88, indicating that the public participation in private PEB is of a high frequency, while public PEB is of a low frequency. This result is similar to the conclusions of previous study based on the data of *CGSS 2003* (Wenjuan, 2008). Compared with the data of *CGSS 2003*, private PEB has significantly improved. For example, the proportion of people "often bringing shopping bags when purchasing daily necessities" increases from 22.7% to 55.2%. The proportion of public PEB has declined. For example, the proportion of people "never participating in complaints and appeals to solve environmental problems" rises from 82.3% to 87.5%. Overall, China's PEB is basically consistent with the development trend of international PEB, that is, private

PEB is improving, while public PEB shows a downward trend (Dalton, 2015).

4.2 Hierarchical Model Results

4.2.1 Hierarchical Model Results for Private Pro-Environment Behavior

Table 4 gives models that influence private PEB. Model 1 is a null model to measure the contributions of private PEB at the individual and city levels. The results of intra-group correlation coefficient (ICC = 70.444/(70.444 + 442.539) = 0.137) suggest that nearly 13.7% of the difference in private PEB among respondents is attributable to city-level factors regardless of any explanatory variable. This demonstrates that the differences in city-level factors should not be ignored in analyzing private PEB, that is, it is necessary to adopt a hierarchical model.

Model 2 adds six individual level variables on the basis of the null model, and these six variables exert significant effects on private PEB. Consistent with previous studies, it is found that years of education have a significantly positive impact on private PEB (b = 1.022, p < 0.01). Women have a higher PEB index than men (b = -4.852, p < 0.01). The private PEB index of the elderly is higher than that of the young (b = 0.144, p < 0.01). The index of environmental pollution perceived by respondents contributes to their private PEB index (b = 0.05, p < 0.05). Among the prediction variables, SI has a significantly positive impact on private PEB. A higher frequency of SI with friends raises the possibility of private PEB (b = 0.756, p < 0.01). EK also positively influences private PEB. Therefore, a well-acquired EK can promote private PEB (b = 1.802, p < 0.01).

All the city-level variables were added to Model 3 to systematically discuss the impact of urbanization on private PEB, including POU, ECU, SPU, SOU and environmental pollution index. The results show that ECU notably influences private PEB (b = 17.621, p < 0.1). Every 1-unit increase in ECU can cause 17.621-unit rise in the index of private PEB, consistent with Hypothesis 2a. SPU also plays a significant role in private PEB (b = -15.754, p < 0.1). When SPU rises by 1 unit, the private PEB index decreases by 15.754 units, consistent with Hypothesis 3a. POU and SOU exert little influence on private PEB (p > 0.1), which fails to support Hypotheses 1a and 4a. Environmental pollution, which serves as a control variable, has a significantly positive impact on private PEB (b = 23.100, p < 0.05), consistent with the pollution-driving hypothesis. Although the variance at individual level in Model 3 is similar to that in Model 2, the variance at city level decreases from 71.061 to 54.601. The results disclose that the city-level variables can explain the 23% difference in private PEB among different cities.

Model 4 is the final model to examine the moderation of urbanization between individual factors and private PEB. The results reveal that SOU positively affects the relationship between SI and private PEB (b = 5.116, p < 0.05), that is, SOU enhances the positive impact of SI on private PEB. However, ECU negatively affects the relationship between SI and private PEB (b = -7.476, p < 0.01). Four dimensions of urbanization do not play a significant moderating role in the relationship between EK and

TABLE 5	Impact of	urbanization	on	public	PFB.	the	results	of HI Ms
I ADEE V	inipact of	andanzation	011	public	· LD.	110	roouito	

	Model 1	Model 2	Model 3	Model 4
	<u> </u>			
Fixed Effect	Coet.	Coet.	Coet.	Coet.
Intercept	9.176***	9.171***	7.371***	9.275***
Individual Level				
SI		1.255***	1.255***	0.92***
EK		0.394	0.394	0.791***
EPPI		0.024*	0.024*	0.027**
Age		0.016	0.016	0.021
Gender		-0.166	-0.166	-0.292
Education		0.676***	0.676***	0.630***
City Level				
POU			5.573	5.569
ECU			4.894	4.879
SPU			-11.084*	-11.398*
SOU			11.084	11.076
EPI			-1.117	-1.100
Cross-Level Interaction	۱			
POU × SI				-2.560
ECU × SI				4.590*
SPU × SI				-0.716
SOU × SI				0.437
POU × EK				1.572
ECU × EK				-2.418
SPU × EK				1.932
SOU × EK				-3.845**
Random Effect	Variance	Variance	Variance	Variance
Level 2 effect	28.979***	28.088***	26.336***	26.385***
Level 1 effect	264.966	251.875	251.930	248.491

Note: *p < 0.1, **p < 0.05, ***p < 0.01; City(N) = 75, Individual(n) = 6,728.

private PEB. Overall, the results partially support Hypothesis 5a, but do not support Hypothesis 5b.

4.2.2 Results of Hierarchical Models for Public Pro-Environment Behavior

Table 5 gives models that affect public PEB. Model 1 is the null model whose results of intra-group correlation coefficient (ICC = 28.979/(28.979 + 264.966) = 0.099) suggest that around 9.9% differences in public PEB among respondents can be explained by differences among cities.

According to the results of Model 2, at the individual level, similar to the results of private PEB, years of education, perceived environmental pollution and SI still play a significant role. Gender, age and EK no longer play an important role in PEB.

All the city-level variables are included in Model 3. The results indicate that SPU still exerts a remarkably negative impact on public PEB (b = -11.084, p < 0.1), in line with Hypothesis 3b. POU, ECU, SOU and environmental pollution index do not have a significant role in public environmental protection, which fails to support Hypotheses 1b, 2b and 4b.

Model 4 includes moderating variables to examine the moderating effect of urbanization on individual factors and public PEB. The results indicate that ECU positively influences the relationship between SI and public PEB (b = 4.590, p < 0.1), that is, ECU enhances the positive impact of SI on public PEB. SOU negatively moderates the relationship between EK and public PEB (b = -3.845, p < 0.05). Other dimensions of urbanization cannot significantly moderate the relationship

between individual factors and public PEB. In general, the research results are partially in line with Hypothesis 5a yet do not support Hypothesis 5b.

5 DISCUSSION

This study aims to explore the influence of urbanization in China on PEB. Its specific goal is to examine the impacts of multidimensions of urbanization (i.e., POU, ECU, SPU and SOU) on PEB, and to explore the cross-level moderating effect of urbanization on the relationship between individual factors and PEB. Therefore, based on the model estimation, hypotheses of the relationship between urbanization and PEB were established and tested. The test results are given in **Table 6**. It is found that urbanization directly affects PEB and moderates the relationship between individual factors and PEB.

First, ECU significantly promotes private PEB, but its impact on public PEB is no significant. The results support Hypothesis 2a, yet they are in inconformity with Hypothesis 2b. Specifically, improving ECU can significantly promote private PEB. This is consistent with the affluence hypothesis. Specifically, Diekmann and Franzen (1999) held that "economic development can promote environmental concern and PEB". The reason is that economic growth not only leads to higher requirements for the environment but also strengthens the public's ability to improve environmental quality. Studies have shown that economic growth and urbanization can promote each other. Urbanization promotes economic growth through the accumulation of physical capital, human capital and knowledge capital; economic growth brings population agglomeration, which leads to large-scale integration of economic activities and spatial population agglomeration, thus improving urbanization (Liang and Yang, 2019). Therefore, strengthening the positive interaction between urbanization and economic growth and maintaining sustainable and healthy economic development are of great significance to promote private PEB.

ECU rarely affects public PEB. It may be because this study did not take institutional factors into consideration, such as political opportunity structure, environmental protection system and political power distribution, which may limit people's participation in public PEB (Hadler and Haller, 2011). For example, with respect to the participation of environmental organizations in environmental governance, most studies have confirmed that environmental organizations in western countries actively promote PEB (Saunders, 2007; Grant and Vasi, 2017). However, Hong and Park (2018) found that the number of environmental non-governmental organizations (NGOs) in China has an insignificant impact on public PEB. According to Lei (2011), nearly half of China's environmental NGOs are organized by the government. They are created and supported by the government, which limits their independent development. Thus, although ECU has contributed to the growth in the number of environmental NGOs in China, they still differ from the traditional western environmental NGOs. Therefore, they cannot efficiently perform their functions, resulting in

TABLE 6 | Hypothesis test result.

Test results
Rejected
Rejected
Adopted
Partly adopted
Adopted
Adopted
Rejected
Rejected
Partly adopted
Rejected

inefficient public participation in environmental governance and a weak influence on environmental policies.

Second, SPU has a significantly negative impact on private PEB and public PEB, which supports Hypotheses 3a and 3b. The formation of urban carriers with modern civilization characteristics and the improvement of infrastructure such as traffic conditions are typical characteristics of SPU. Although China has dedicated itself to improving infrastructure and increasing investment in infrastructure during urbanization, it is still faced with problems such as unsatisfying quality of infrastructure, extensive operation and management. These problems may constrain PEB (Fan et al., 2018). In recent years, an increasing number of scholars have discussed that infrastructure availability is more important than the number of facilities (Bernstad, 2014; Fan et al., 2018). Martin et al. (2006) disclosed that household waste recycling practice is hindered by its inconvenience. In China, most cities, even developed cities including Beijing and Shanghai, neglect improving the availability of infrastructure, such as the location of garbage bins. In terms of public transport facilities, "in most cities of China, publictransport-motorized trips account for less than 40% of the total trips, displaying a far cry from similar foreign cities". One important reason is that bad experience weakens the public's motivation for green travel by public transport, such as "long waiting time, low driving speed, poor riding environment and inconvenient transfer" (MOT China, 2016). Therefore, during urbanization, more attention should be paid to the availability of infrastructure and operation management services, so as to promote an active participation of the public in PEB.

Third, SOU has a positive yet slight impact on both private PEB and public PEB, which fails to support Hypotheses 4a and 4b. A possible explanation is as follows: For one thing, SOU can promote PEB by improving public services (Fan et al., 2018); however, as China's urban public services have been planned according to the household population for a long time, some cities, especially big cities, are facing higher public service pressure. For example, people choose to queue all night for the education and housing resources in shortage. Therefore, the insufficient supply of urban public services may weaken its role in promoting PEB. For another, according to the post materialism theory, with the prosperity of society and the improvement of living quality, the public's priority values have changed from the original "materialism" to "post materialism", paying more attention to self-value and ecological environment. However, as proposed by Inglehart, China is currently in a period of great transformation. Although China has completed the transformation from a stage of food and clothing crisis, social instability, poverty and backwardness to a relatively prosperous stage in just a few decades, it has not yet entered the post materialist stage (Inglehart, 2013). Therefore, in the context of insufficient SOU development, the Chinese public may still be more concerned about the lower-level needs that are directly related to their own interests.

Fourth, POU has no significant effect on both private PEB and public PEB, which fails to support Hypotheses 1a and 1b. A previous study displayed the same results (Franzen and Meyer, 2010). In the cross-country study involving 26 countries, Franzen and Meyer (2010) found that a country's population density or the proportion of urban population did not significantly influence the public's concern about the environment. A possible explanation is as follows: First, most studies believe that population density and population agglomeration can cause serious environmental pressure and pollution, so people in densely populated cities may be more concerned about the environment and are more willing to participate in PEB. However, through years of longitudinal researches on China's cities, Glaeser and Kahn (2010) found that urban population agglomeration is not bound to bring more serious environmental pollution. For example, a dense subway network supported by a high population density can reduce the number of private car trips instead. In other words, POU will not necessarily cause higher environmental pressure. Direct confrontation with environmental issues is not the only factor that causes greater pro-environment awareness and willingness to participate in environmental protection. Besides, from the perspective of population migration, although some study held the view that migrants can improve their PEB through learning (Pfeffer and Stycos, 2002), other study found that migrants have less impact on the environment than social institutional factors such as economic, political structures and religious beliefs (Ng, 1998). This means social factors including economic and political structures may weaken the impact of population factors on PEB in the process of urbanization.

Fifth, it is found that the level of urbanization significantly moderates the relationship between SI and PEB, while its moderating effect on EK and PEB is insignificant. SI significantly improves both private and public PEB, which is consistent with the conclusions of previous studies (Franzen and Meyer, 2010; Wang and Han, 2016; Zheng et al., 2019). According to those studies, during SI, the public can moderate their PEB by observing the PEB of others. In addition, the public can actively moderate their PEB by acquiring EK. Compared with cities with lower urbanization levels, cities with higher urbanization levels provide more communication and interaction opportunities between different occupational groups, which is more conducive to promoting SI of the public. Therefore, the level of urbanization can moderate the relationship between SI and PEB of the public.

ECU strengthens the positive relationship between SI and public PEB, that is, ECU can indirectly affect public PEB through SI. Munro (2014) found that social capital is related to greater environmental concern and causes a greater possibility of participation in collective action. In environmental practice, because of the cost and risk of resistant public PEB (Wiltfang and McAdam, 1991), people may not have enough economic capacity and risk resistance to support public PEB, even if they are more willing to accept the views from friends and more likely to learn and imitate their behavior. However, in this case, a higher level of ECU plays a catalytic role (Gillham, 2008). In addition, ECU weakens the positive relationship between SI and private PEB. The positive effect of ECU is mainly reflected in its direct impact on private PEB. In other words, when ECU improves, the public will pay attention to environmental issues and actively participate in private PEB, which is consistent with the affluence hypothesis (Diekmann and Franzen, 1999). Therefore, ECU may weaken SI, which serves as a prerequisite for private PEB. Next, SOU can strengthen the relationship between SI and private PEB. This is also in line with Maslow's hierarchy of needs theory, which holds that when people's physiological needs and safety needs are basically met, social needs become a strong motivation, and people begin to desire to establish emotional connections or relationships with others (Maslow, 1981). In a society with a higher level of urbanization, people have stronger social needs and more frequent SI, thus promoting private PEB. EK actively promotes private PEB, which agrees with previous studies (Liu et al., 2020; Amoah and Addoah, 2021). This means that the publicity and education ΕK of remain important means in environmental protection. However, the moderating effect of urbanization on the relationship between EK and PEB has not been confirmed in this study. A possible explanation is that although urbanization has improved the overall education level, environmental education should be further improved (Fan et al., 2018). Therefore, in urbanization development, it is necessary to improve the publicity and education of EK to promote the transformation of EK to PEB (Fan et al., 2018).

6 CONCLUSION

6.1 Research Conclusion

Many studies have focused on the pressure brought to resources and environment by urbanization, but few have explored the positive effect of urbanization on reducing environmental pressure from the perspective of urbanization affecting PEB. This research reveals that urbanization not only directly affects PEB, but also moderates the relationship between individual characteristics and PEB. The impacts of urbanization with different dimensions on PEB differ. First, ECU can promote private PEB, indicating that economic development increases the public demand for environment and the ability to improve the environment. However, due to institutional constraints, ECU exerts no direct influence on public PEB. Second, SOU has a positive yet slight impact on PEB, indicating an insufficient social development in China and an imbalance between supply and demand of urban public services. Third, SPU has a significantly negative impact on PEB. Although China continuously increases investment in urban infrastructure, prominent problems may constrain PEB, such as unsatisfying quality of infrastructure and extensive operation and management. In addition, consistent with previous studies, it is found that POU has no significant impact on PEB. Finally, SI actively promotes PEB; this promotion effect can be positively moderated by SOU and ECU. In the process of SI, people can moderate and promote their PEB by observing the PEB of others and acquiring EK. Compared with people in cities with relatively lower SOU levels, people in cities with higher SOU levels correspond to stronger social desires and more frequent SI, which promote the private PEB. Compared with people in cities with relatively lower ECU levels, people in cities with higher ECU levels have more social capital accumulation and can handle risks better, so they are more likely to participate in costly and risky public PEB.

6.2 Practical Implications

These findings are of important theoretical guiding significance for capturing the complex relationship between urbanization and PEB, and of certain practical reference value for promoting the specific policy design of PEB. In practice, first, a green and sustainable ECU should be promoted, because it is not only an important prerequisite for building an environment-friendly society (Wang and Han, 2016), but also the basis for promoting PEB. Second, SOU should be strengthened. PEB can be promoted by improving public services and the life quality of residents. Third, attention should be given to the smart integration of the availability of urban infrastructure and residents' life (Ma et al., 2018), hereby promoting residents to have a green lifestyle. Finally, SI can be improved by enriching community activities, so as to promote public participation in environmental protection (Zhu et al., 2021).

6.3 Limitations and Future Studies

Though this study expands the scope of studies on the relationship between urbanization and PEB, certain limitations remain in this study. First, though this study explores the direct and indirect effects of urbanization on PEB, urbanization is actually a complex systematic project, which influences PEB from various aspects. Hence, future researches should consider the impacts of institutional factors on PEB, such as social institutional

arrangements, social governance framework and effectiveness. These factors may further reveal the internal causes and behavioral mechanisms of the differences in private PEB and public PEB at different urbanization levels. Second, in terms of research methods, the data of independent variables and dependent variables in this study were taken at different times, which are more favorable for testing the effect of independent variables on dependent variables than cross-sectional data. This is conducive to the test of causality. However, it still lacks longitudinal studies in the same area, which fails to display the dynamic effect of urbanization on PEB. Therefore, future researches can further explore the impact of urbanization on PEB from the perspective of vertical design by combining the continuous multi-year indexes of databases. Third, with respect to the measurement index, the limitation of data sources may lead to deficiencies in the established urbanization index system. In the future study, a more comprehensive assessment of the level of urbanization will be conducted, especially a more accurate assessment of its specific dimensions.

REFERENCES

- Agency, X. N. (2021). The Fourteenth Five-Year Plan. Available online: http:// www.gov.cn/xinwen/2021-03/13/content_5592681.htm (accessed on September 19, 2021).
- Ahmad, N., Ullah, Z., Arshad, M. Z., Kamran, H. W., Scholz, M., and Han, H. (2021). Relationship between Corporate Social Responsibility at the Micro-Level and Environmental Performance: The Mediating Role of Employee Proenvironmental Behavior and the Moderating Role of Gender. Sustain. Prod. Consum. 27, 1138–1148. doi:10.1016/j.spc.2021.02.034
- Amoah, A., and Addoah, T. (2021). Does Environmental Knowledge Drive Pro-Environmental Behaviour in Developing Countries? Evidence from Households in Ghana. *Environ. Dev. Sustain* 23 (2), 2719–2738. doi:10. 1007/s10668-020-00698-x
- Ates, H. (2020). Merging Theory of Planned Behavior and Value Identity Personal Norm Model to Explain Pro-Environmental Behaviors. Sustain. Prod. Consum. 24, 169–180. doi:10.1016/j.spc.2020.07.006
- Azucena Vicente-Molina, M., Fernández-Sáinz, A., and Izagirre-Olaizola, J. (2013). Environmental Knowledge and Other Variables Affecting Pro-Environmental Behaviour: Comparison of University Students from Emerging and Advanced Countries. J. Clean. Prod. 61, 130–138. doi:10.1016/j.jclepro.2013.05.015
- Bernstad, A. (2014). Household Food Waste Separation Behavior and the Importance of Convenience. Waste Manag. 34 (7), 1317–1323. doi:10.1016/ j.wasman.2014.03.013
- Cao, Y., Heng, C. K., and Fung, J. C. (2019). Using Walk-Along Interviews to Identify Environmental Factors Influencing Older Adults' Out-Of-Home Behaviors in a High-Rise, High-Density Neighborhood. Int. J. Environ. Res. Public Health 16 (21), 4251. doi:10.3390/ijerph16214251
- Chen, J., Wang, L., and Li, Y. (2020). Research on the Impact of Multi-Dimensional Urbanization on China's Carbon Emissions under the Background of COP21. *J. Environ. Manag.* 273, 111123. doi:10.1016/j.jenvman.2020.111123
- Chen, X., Peterson, M. N., Hull, V., Lu, C., Hong, D., and Liu, J. (2013). How Perceived Exposure to Environmental Harm Influences Environmental Behavior in Urban China. *Ambio* 42 (1), 52–60. doi:10.1007/s13280-012-0335-9
- Conway, J., and Adams, B. (1977). The Social Effects of Living off the Ground. Habitat Int. 2 (5-6), 595–614. doi:10.1016/0197-3975(77)90032-7
- Dalton, R. J. (2015). Waxing or Waning? The Changing Patterns of Environmental Activism. Environ. Polit. 24 (4), 530–552. doi:10.1080/09644016.2015.1023576
- Davis, J. C., and Henderson, J. V. (2003). Evidence on the Political Economy of the Urbanization Process. J. Urban Econ. 53 (1), 98–125. doi:10.1016/s0094-1190(02)00504-1

DATA AVAILABILITY STATEMENT

Publicly available datasets were analyzed in this study. This data can be found here: http://cgss.ruc.edu.cn/.

AUTHOR CONTRIBUTIONS

WQ conceived and wrote the first manuscript. YW conceptualized and supervised the study. JH and CL acquired the data and conducted the data analysis. WQ and YZ revised the manuscript. All authors read and approved the final manuscript.

FUNDING

This study is supported by the Graduate Research & Practice Innovation Program of Jiangsu Province (KYCX21_2099).

- Derksen, L., and Gartrell, J. (1993). The Social Context of Recycling. Am. Sociol. Rev. 58, 434–442. doi:10.2307/2095910
- Diekmann, A., and Franzen, A. (1999). The Wealth of Nations and Environmental Concern. Environ. Behav. 31 (4), 540–549. doi:10.1177/00139169921972227
- Ding, C., and Zhao, X. (2011). Assessment of Urban Spatial-Growth Patterns in China during Rapid Urbanization. *Chin. Econ.* 44 (1), 46–71. doi:10.2753/ CES1097-1475440104
- Ertz, M., Karakas, F., and Sarigöllü, E. (2016). Exploring Pro-Environmental Behaviors of Consumers: An Analysis of Contextual Factors, Attitude, and Behaviors. J. Bus. Res. 69 (10), 3971–3980. doi:10.1016/j.jbusres.2016. 06.010
- Etminani-Ghasrodashti, R., Paydar, M., and Ardeshiri, A. (2018). Recreational Cycling in a Coastal City: Investigating Lifestyle, Attitudes and Built Environment in Cycling Behavior. *Sustain. Cities Soc.* 39, 241–251. doi:10. 1016/j.scs.2018.02.037
- Fan, B., Yang, W., and Han, T. (2018). Impact of Basic Public Service Level on Pro-Environmental Behavior in China. *Int. Sociol.* 33 (6), 738–760. doi:10.1177/ 0268580918792778
- Fan, M., Khalique, A., Qalati, S. A., Gillal, F. G., and Gillal, R. G. (2021). Antecedents of Sustainable E-Waste Disposal Behavior: the Moderating Role of Gender. *Environ. Sci. Pollut. Res.* 29, 20878–20891. doi:10.1007/s11356-021-17275-y
- Franzen, A., and Meyer, R. (2010). Environmental Attitudes in Cross-National Perspective: A Multilevel Analysis of the ISSP 1993 and 2000. *Eur. Sociol. Rev.* 26 (2), 219–234. doi:10.1093/esr/jcp018
- Gelissen, J. (2007). Explaining Popular Support for Environmental Protection. Environ. Behav. 39 (3), 392-415. doi:10.1177/0013916506292014
- Gifford, R., and Nilsson, A. (2014). Personal and Social Factors that Influence Pro-Environmental Concern and Behaviour: A Review. Int. J. Psychol. 49 (3), 141–157. doi:10.1002/ijop.12034
- Gillham, P. F. (2008). Participation in the Environmental Movement: Analysis of the European Union. *Int. Sociol.* 23 (1), 67–93. doi:10. 1177/0268580907084386
- Glaeser, E. L., and Kahn, M. E. (2010). The Greenness of Cities: Carbon Dioxide Emissions and Urban Development. J. Urban Econ. 67 (3), 404–418. doi:10. 1016/j.jue.2009.11.006
- Grant, D., and Vasi, I. B. (2017). Civil Society in an Age of Environmental Accountability: How Local Environmental Nongovernmental Organizations Reduce U.S. Power Plants' Carbon Dioxide Emissions. Sociol. Forum 32 (1), 94–115. doi:10.1111/socf.12318
- Guagnano, G. A., Stern, P. C., and Dietz, T. (1995). Influences on Attitude-Behavior Relationships: A Natural Experiment with Curbside Recycling. *Environ. Behav.* 27 (5), 699–718. doi:10.1177/0013916595275005

- Hadler, M., and Haller, M. (2011). Global Activism and Nationally Driven Recycling: The Influence of World Society and National Contexts on Public and Private Environmental Behavior. *Int. Sociol.* 26 (3), 315–345. doi:10.1177/ 0268580910392258
- Han, F., Xie, R., Lu, Y., Fang, J., and Liu, Y. (2018). The Effects of Urban Agglomeration Economies on Carbon Emissions: Evidence from Chinese Cities. J. Clean. Prod. 172, 1096–1110. doi:10.1016/j.jclepro.2017.09.273
- Hao, F., and Song, L. (2020). Environmental Concern in China: A Multilevel Analysis. Chin. Sociol. Rev. 52 (1), 1–26. doi:10.1080/21620555.2019.1654367
- Hong, Z., and Park, I. (2018). The Effects of Regional Characteristics and Policies on Individual Pro-Environmental Behavior in China. Sustainability 10 (10), 3586. doi:10.3390/su10103586
- Hunter, L. M. (2000). A Comparison of the Environmental Attitudes, Concern, and Behaviors of Native-Born and Foreign-Born U.S. Residents. *Popul. Environ.* 21 (6), 565–580. doi:10.1007/bf02436772
- Hunter, L. M., Hatch, A., and Johnson, A. (2004). Cross-National Gender Variation in Environmental Behaviors. Soc. Sci. Q 85 (3), 677–694. doi:10. 1111/j.0038-4941.2004.00239.x
- Hyslop, N. P. (2009). Impaired Visibility: The Air Pollution People See. Atmos. Environ. 43 (1), 182–195. doi:10.1016/j.atmosenv.2008.09.067
- Inglehart, R. F. (2013). China Has Not yet Entered the Stage of Post-materialist Values. *People's Trib.* (27), 50–51. doi:10.16619/j.cnki.rmlt.2013.27.007
- Inglehart, R. (1995). Public Support for Environmental Protection: Objective Problems and Subjective Values in 43 Societies. PS Political Sci. Polit. 28 (1), 57–72. doi:10.2307/420583
- Jin, G., Deng, X., Zhao, X., Guo, B., and Yang, J. (2018). Spatiotemporal Patterns in Urbanization Efficiency within the Yangtze River Economic Belt between 2005 and 2014. J. Geogr. Sci. 28 (8), 1113–1126. doi:10.1007/s11442-018-1545-2
- Jin, G., Shi, X., He, D., Guo, B., Li, Z., and Shi, X. (2020). Designing a Spatial Pattern to Rebalance the Orientation of Development and Protection in Wuhan. J. Geogr. Sci. 30 (4), 569–582. doi:10.1007/s11442-020-1743-6
- Klein, K. J., and Kozlowski, S. W. J. (2000). From Micro to Meso: Critical Steps in Conceptualizing and Conducting Multilevel Research. Organ. Res. Methods 3, 211–236. doi:10.1177/109442810033001
- Kuang, B., Lu, X., Han, J., Fan, X., and Zuo, J. (2020). How Urbanization Influence Urban Land Consumption Intensity: Evidence from China. *Habitat Int.* 100, 102103. doi:10.1016/j.habitatint.2019.102103
- Kuang, Y., and Lin, B. (2021). Public Participation and City Sustainability: Evidence from Urban Garbage Classification in China. Sustain. Cities Soc. 67, 102741. doi:10.1016/j.scs.2021.102741
- Lei, X. (2011). China's Environmental Activism in the Age of Globalization. *Asian Polit. Policy* 3 (2), 207–224. doi:10.1111/j.1943-0787.2011.01256.x
- Levine, D. S., and Strube, M. J. (2012). Environmental Attitudes, Knowledge, Intentions and Behaviors Among College Students. J. Soc. Psychol. 152 (3), 308–326. doi:10.1080/00224545.2011.604363
- Liang, W., and Yang, M. (2019). Urbanization, Economic Growth and Environmental Pollution: Evidence from China. Sustain. Comput. Inf. Syst. 21, 1–9. doi:10.1016/j.suscom.2018.11.007
- Lin, S., Sun, J., Marinova, D., and Zhao, D. (2017). Effects of Population and Land Urbanization on China's Environmental Impact: Empirical Analysis Based on the Extended STIRPAT Model. Sustainability 9 (5), 825. doi:10.3390/su9050825
- Liu, H., and Lei, J. (2018). The Impacts of Urbanization on Chinese Households' Energy Consumption: An Energy Input-Output Analysis. J. Renew. Sustain. Energy 10 (1), 015903. doi:10.1063/1.5020077
- Liu, P., Teng, M., and Han, C. (2020). How Does Environmental Knowledge Translate into Pro-Environmental Behaviors? The Mediating Role of Environmental Attitudes and Behavioral Intentions. *Sci. Total Environ.* 728, 138126. doi:10.1016/j.scitotenv.2020.138126
- Liu, Y., Huang, B., Wang, R., Feng, Z., Liu, Y., and Li, Z. (2019). Exploring the Association between Urbanisation and Self-Rated Health of Older Adults in China: Evidence from a National Population Sample Survey. *BMJ Open* 9 (6), e029176. doi:10.1136/bmjopen-2019-029176
- Ma, J., Hipel, K. W., Hanson, M. L., Cai, X., and Liu, Y. (2018). An Analysis of Influencing Factors on Municipal Solid Waste Source-Separated Collection Behavior in Guilin, China by Using the Theory of Planned Behavior. Sustain. Cities Soc. 37, 336–343. doi:10.1016/j.scs.2017.11.037

- Macias, T., and Williams, K. (2016). Know Your Neighbors, Save the Planet: Social Capital and the Widening Wedge of Pro-Environmental Outcomes. *Environ. Behav.* 48 (3), 391–420. doi:10.1177/0013916514540458
- Markandya, A., Baumol, W. J., and Oates, W. E. (1979). Economics, Environmental Policy, and the Quality of Life. *Econ. J.* 89 (356), 1002–1004. doi:10.2307/ 2231545
- Martin, M., Williams, I. D., and Clark, M. (2006). Social, Cultural and Structural Influences on Household Waste Recycling: A Case Study. *Resour. Conserv. Recycl.* 48 (4), 357–395. doi:10.1016/j.resconrec.2005.09.005
- Maslow, A. H. (1954). Motivation and Personality. New York: Harper.
- McCright, A. M., and Xiao, C. (2014). Gender and Environmental Concern: Insights from Recent Work and for Future Research. Soc. Nat. Resour. 27 (10), 1109–1113. doi:10.1080/08941920.2014.918235
- Meyer, A. (2015). Does Education Increase Pro-Environmental Behavior? Evidence from Europe. *Ecol. Econ.* 116, 108–121. doi:10.1016/j.ecolecon. 2015.04.018
- Mi, L., Sun, Y., Gan, X., Yang, Y., Jia, T., Wang, B., et al. (2021). Predicting Environmental Citizenship Behavior in the Workplace: A New Perspective of Environmental Affective Event. *Sustain. Prod. Consum.* 27, 2037–2046. doi:10. 1016/j.spc.2021.05.006
- Monier, S., Hassan, D., Nichèle, V., and Simioni, M. (2009). Organic Food Consumption Patterns. J. Agric. Food Ind. Organ. 7 (2), 12. doi:10.2202/ 1542-0485.1269
- Montgomery, J. (1998). Making a City: Urbanity, Vitality and Urban Design. J. Urban Des. 3 (1), 93–116. doi:10.1080/13574809808724418
- MOT China (2016). "13th Five-Year Plan" Development Outline of Urban Public Transport. Available online: https://www.mot.gov.cn/xiazaizhongxin/ ziliaoxiazai/201607/P020200709613331794690.pdf (accessed on September 19, 2021).
- Munro, N. (2014). Profiling the Victims: Public Awareness of Pollution-Related Harm in China. J. Contemp. China 23 (86), 314–329. doi:10.1080/10670564. 2013.832532
- National Bureau of Statistics (2021). Bulletin of the Seventh China Census (No.7). Available online: http://www.stats.gov.cn/tjsj/zxfb/202105/t20210510_ 1817183.html (accessed on September 23, 2021).
- Ng, C. F. (1998). Canada as a New Place: The Immigrant's Experience. J. Environ. Psychol. 18 (1), 55–67. doi:10.1006/jevp.1997.0065
- O'Connor, R. T., Lerman, D. C., Fritz, J. N., and Hodde, H. B. (2010). Effects of Number and Location of Bins on Plastic Recycling at a University. J. Appl. Behav. Analysis 43 (4), 711–715. doi:10.1901/jaba.2010.43-711
- Pfeffer, M. J., and Stycos, J. M. (2002). Immigrant Environmental Behaviors in New York City. Soc. Sci. Q 83 (1), 64–81. doi:10.1111/1540-6237.00071
- Pisano, I., and Lubell, M. (2017). Environmental Behavior in Cross-National Perspective: A Multilevel Analysis of 30 Countries. *Environ. Behav.* 49 (1), 31–58. doi:10.1177/0013916515600494
- Ren, F., and Yu, X. (2021). Coupling Analysis of Urbanization and Ecological Total Factor Energy Efficiency -- A Case Study from Hebei Province in China. Sustain. Cities Soc. 74, 103183. doi:10.1016/j.scs.2021.103183
- Ruxin, W. (2017). Urbanization, Basic Public Service Supply and Subjective Well-Being: An Empirical Study on 56 Cities in CGSS(2010). *Popul. Dev.* 23 (03), 37–48+67.
- Sato, Y., and Zenou, Y. (2015). How Urbanization Affect Employment and Social Interactions. Eur. Econ. Rev. 75, 131–155. doi:10.1016/j.euroecorev.2015.01.011
- Saunders, C. (2007). The National and the Local: Relationships Among Environmental Movement Organisations in London. *Environ. Polit.* 16 (5), 742–764. doi:10.1080/09644010701634083
- Shang, J., Wang, Z., Li, L., Chen, Y., and Li, P. (2018). A Study on the Correlation between Technology Innovation and the New-Type Urbanization in Shaanxi Province. *Technol. Forecast. Soc. Change* 135, 266–273. doi:10.1016/j.techfore. 2018.04.029
- Shen, J., and Saijo, T. (2008). Reexamining the Relations between Socio-Demographic Characteristics and Individual Environmental Concern: Evidence from Shanghai Data. J. Environ. Psychol. 28 (1), 42–50. doi:10. 1016/j.jenvp.2007.10.003
- Steg, L., and Vlek, C. (2009). Encouraging Pro-Environmental Behaviour: An Integrative Review and Research Agenda. J. Environ. Psychol. 29 (3), 309–317. doi:10.1016/j.jenvp.2008.10.004

- Stern, P. C. (2000). New Environmental Theories: Toward a Coherent Theory of Environmentally Significant Behavior. J. Soc. Isssues 56 (3), 407–424. doi:10. 1111/0022-4537.00175
- Stern, P. C., and Paul, C. (2000). New Environmental Theories: Toward a Coherent Theory of Environmentally Significant Behavior. J. Soc. Isssues 56 (3), 407–424. doi:10.1111/0022-4537.00175
- Strapko, N., Hempel, L., MacIlroy, K., and Smith, K. (2016). Gender Differences in Environmental Concern: Reevaluating Gender Socialization. Soc. Nat. Resour. 29 (9), 1015–1031. doi:10.1080/08941920.2016.1138563
- Summers, N., and VanHeuvelen, T. (2017). Heterogeneity in the Relationship between Country-Level Affluence and Environmental Concern. Soc. Forces 96 (1), 329–360. doi:10.1093/sf/sox038
- Sun, Y., Du, Y., Wang, Y., and Zhuang, L. (2017). Examining Associations of Environmental Characteristics with Recreational Cycling Behaviour by Street-Level Strava Data. Int. J. Environ. Res. Public Health 14 (6), 644. doi:10.3390/ ijerph14060644
- Timlett, R. E., and Williams, I. D. (2009). The Impact of Transient Populations on Recycling Behaviour in a Densely Populated Urban Environment. *Resour. Conservation Recycl.* 53 (9), 498–506. doi:10.1016/j.resconrec.2009.03.010
- Tindall, D. B., Davies, S., and Mauboulès, C. (2003). Activism and Conservation Behavior in an Environmental Movement: The Contradictory Effects of Gender. Soc. Nat. Resour. 16 (10), 909–932. doi:10.1080/716100620
- Tsui, A. S., Ashford, S. J., Clair, L. S., and Xin, K. R. (1995). Dealing with Discrepant Expectations: Response Strategies and Managerial Effectiveness. *Acad. Manag.* J. 38 (6), 1515–1543. doi:10.5465/256842
- Wang, M.-S., and Chien, H.-T. (1998). Environmental Behaviour Analysis of High-Rise Building Areas in Taiwan. *Build. Environ.* 34 (1), 85–93. doi:10.1016/ s0360-1323(97)00070-x
- Wang, R., Lu, Y., Wu, X., Liu, Y., and Yao, Y. (2020). Relationship between Eye-Level Greenness and Cycling Frequency Around Metro Stations in Shenzhen, China: A Big Data Approach. Sustain. Cities Soc. 59, 102201. doi:10.1016/j.scs.2020.102201
- Wang, Y., Hao, F., and Liu, Y. (2021). Pro-Environmental Behavior in an Aging World: Evidence from 31 Countries. *Int. J. Environ. Res. Public Health* 18 (4), 1748. doi:10.3390/ijerph18041748
- Wang, Y. J., and Han, D. L. (2016). Economic Development, Environmental Pollution and Public Pro-Environmental Behavior——A Multi-level Analysis of the 2013 Chinese General Social Survey. J. Renmin Univ. China 2, 79–92.
- Welsch, H., and Kühling, J. (2017). Pan-European patterns of environmental concern: the role of proximity and international integration. J. Environ. Stud. Sci. 7 (4), 473–489. doi:10.1007/s13412-017-0441-x
- Wendling, Z. A., Emerson, J. W., de Sherbinin, A., Esty, D. C., et al. (2020). 2020 Environmental Performance Index. New Haven, CT: Yale Center for Environmental Law & Policy. epi.yale.edu.
- Wenjuan, G. (2008). A Gender-Based Comparison of Chinese Urban Residents' Environmentally Friendly Behaviors. Collect. Women's Stud. 6 (06), 11–17.
- Wiltfang, G. L., and McAdam, D. (1991). The Costs and Risks of Social Activism: A Study of Sanctuary Movement Activism. Soc. Forces 69 (4), 987–1010. doi:10. 2307/2579299
- Wu, L., Zhu, Y., and Zhai, J. (2022). Understanding Waste Management Behavior Among University Students in China: Environmental Knowledge, Personal

Norms, and the Theory of Planned Behavior. Front. Psychol. 12, 13. doi:10. 3389/fpsyg.2021.771723

- Xiao, C., and Dunlap, R. E. (2007). Validating a Comprehensive Model of Environmental Concern Cross-Nationally: A U.S.-Canadian Comparison. Soc. Sci. Q 88 (2), 471–493. doi:10.1111/j.1540-6237.2007.00467.x
- Xiao, C., and Hong, D. (2018). Gender Differences in Environmental Behaviors Among the Chinese Public: Model of Mediation and Moderation. *Environ. Behav.* 50 (9), 975–996. doi:10.1177/0013916517723126
- Xiao, C., and McCright, A. M. (2014). A Test of the Biographical Availability Argument for Gender Differences in Environmental Behaviors. *Environ. Behav.* 46 (2), 241–263. doi:10.1177/0013916512453991
- Yao, F., Zhu, H. S., and Wang, M. J. (2021). The Impact of Multiple Dimensions of Urbanization on CO2 Emissions: A Spatial and Threshold Analysis of Panel Data on China's Prefecture-Level Cities. *Sustain. Cities Soc.* 73, 103113. doi:10. 1016/j.scs.2021.103113
- Yu, B. (2021). Ecological effects of new-type urbanization in China. Renew. Sustain. Energy Rev. 135, 110239. doi:10.1016/j.rser.2020.110239
- Zhang, Y., Liu, Y., and Zhai, K. (2021b). Identifying the Predictors of Community Acceptance of Waste Incineration Plants in Urban China: A Qualitative Analysis from a Public Perspective. *Int. J. Environ. Res. Public Health* 18 (19), 10189. doi:10.3390/ijerph181910189
- Zhang, Y., Wang, Y., Ahmad, A. B., Shah, A. A., and Qing, W. (2021a). How Do Individual-Level Characteristics Influence Cross-Domain Risk Perceptions Among Chinese Urban Residents? SAGE Open 11 (2), 21582440211003570. doi:10.1177/21582440211003570
- Zheng, J., Yang, M., Xu, M., Zhao, C., and Shao, C. (2019). An Empirical Study of the Impact of Social Interaction on Public Pro-Environmental Behavior. *Int. J. Environ. Res. Public Health* 16 (22), 4405. doi:10.3390/ ijerph16224405
- Zhu, Y., Wang, Y., and Liu, Z. (2021). How Does Social Interaction Affect Pro-Environmental Behaviors in China? The Mediation Role of Conformity. *Front. Environ. Sci.* 9, 12. doi:10.3389/fenvs.2021.690361

Conflict of Interest: The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Publisher's Note: All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

Copyright © 2022 Qing, Wang, Zhang, Han and Li. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.